Title: Post-traumatic stress disorder following patient assaults among staff members of mental health hospitals: a prospective longitudinal study

Authors:

Dirk Richter (dirk.richter@uni-muenster.de)
Klaus Berger (bergerk@uni-muenster.de)

Version: 3 Date: 6 April 2006

Author's response to reviews:

Title and author list have been checked - no changes.
Abstract: Your recommended abstract has been copied into the manuscript.
Underlinings: Have been removed.
Introduction: Has been renamed to Background.
Declaration of interests: Has been changed according your recommendation.
Table title: Has been placed above the table.
Ref. 21: Has not been published yet.